Author’s response to reviews

Title: Identification and quantification of change in Australian illicit drug markets

Authors:

Stuart Gilmour (stuart.gilmour@unsw.edu.au)
Inge Koch (inge@maths.unsw.edu.au)
Louisa Degenhardt (l.degenhardt@unsw.edu.au)
Carolyn Day (cday@nchecr.unsw.edu.au)

Version: 4 Date: 16 June 2006

Author’s response to reviews:

Competing interest
The authors declare that they have no competing interests.

Authors’ contributions
All authors 1) have made substantial contributions to conception and design; 2) have been involved in drafting the manuscript or revising it critically for important intellectual content; and 3) have given final approval of the version to be published.

Acknowledgements
This research was funded by the National Drug Law Enforcement Research Foundation (NDLERF). The NDLERF played no role in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.